Author's response to reviews

Title: The mannosylated extracellular domain of Her2/neu produced in P. pastoris induces protective antitumor immunity.

Authors:

Alexios Dimitriadis (aldimitriadis@pasteur.gr)
Chrysanthi Gontinou (xrisanthe78@yahoo.gr)
Constantin N. Baxevanis (baxevanis@ciic.gr)
Avgi Mamalaki (amamalaki@pasteur.gr)

Version: 4 Date: 29 October 2009

Author's response to reviews: see over
Dear Dr. Bucceri,

We hereby submit for publication a revised version of 1923754115263936. We have addressed all the comments regarding the format of the manuscript. A point-by-point response to all issues is provided below. We have also checked the manuscript for any typographical errors and we have made all the necessary corrections. We thank you very much for your attention to this work and hope that you will find the revised manuscript suitable for publication in BMC cancer.

Yours sincerely,

Dr Avgi Mamalaki
Formatting changes requested

Major revisions

1. Corresponding author – One or more corresponding author(s) must be indicated on the title page, and their email address must be given
   The corresponding author is indicated.

2. Figure legends – Figure legend 4, 6 and 7 contain graphics. These will either need to be included within image as a key, or written in text (e.g. “black square”).
   In figure legends 4, 6 and 7 graphics are written in text.

3. Figures – Please provide a title, no longer than one sentence long for each figure.
   A title for each figure is provided.

Minor revisions

1. Country – Please remember to include the country in the affiliation details.
   The country is provided in the affiliation details.

2. Email address – Please include email addresses for all the authors on the title page.
   Email addresses for all the authors are provided.

3. Please cite figures consistently as either fig or figure in both the main text and the figure legends section.
   All figures are cited as “figure” in the main text and the figure legends section.